Author's response to reviews

Title: Does access to a colorectal cancer screening website and/or a nurse-managed telephone help line provided to patients by their family physician increase fecal occult blood test uptake?: Results from a pragmatic cluster randomized controlled trial

Authors:

Kathleen M Clouston (kclousto@cc.umanitoba.ca)
Alan Katz (Alan_Katz@cpe.umanitoba.ca)
Patricia J Martens (pat_martens@cpe.umanitoba.ca)
Jeff Sisler (jeff.sisler@med.umanitoba.ca)
Donna Turner (Donna.Turner@cancercare.mb.ca)
Michelle Lobchuk (michelle_lobchuk@umanitoba.ca)
Susan McClement (susan.mcclement@cancercare.mb.ca)
Gary Crow (gary_crow@umanitoba.ca)
Cornelius Woelk (cw@cwwiebemedical.ca)

Version: 2 Date: 4 October 2013

Author's response to reviews: see over
Dear Mr. Alfredo Auro V. Evangelista,

RE: Request to check Trial Registration Number (TRN) for submitted research article BMC Cancer submission reference number MS: 6777099101097202

On behalf of the authors of “Does access to a colorectal cancer screening website and/or a nurse-managed telephone help line provided to patients by their family physician increase fecal occult blood test uptake?: Results from a pragmatic cluster randomized controlled trial” (referred to above), I am responding to the email requesting a check of the TRN (incorrectly reported as NCT1026753 in the submitted manuscript). It seems there was one number omitted. **The correct TRN is NCT01026753.** That is, the first digit should have started with a zero (which was omitted).

The trial information can be accessed using the following URL:


Please let me know if satisfies your request and/or if there are any additional questions to facilitate the peer review process of the submitted article.

Thank you very much. I look forward to your response.

Sincerely,

Kathleen Clouston

Dr. Kathleen Clouston
Department of Family Medicine (Research)
CIHR Post-Doctoral Fellow; CIHR/CCMB Primary Care Oncology Research Team
kclousto@cc.umanitoba.ca
204-272-3086